Author’s response to reviews

Title: Evaluation of apoptosis and angiogenesis in ectopic and eutopic stromal cells of patients with endometriosis compared to non-endometriotic controls

Authors:

Ali-Akbar Delbandi (delbandi@yahoo.com)

Mahmoud Mahmoudi (mahmoudim@mums.ac.ir)

Adel Shervin (ads1947@yahoo.com)

Sahel Heidari (heidari_sahel69@yahoo.com)

Roya Kolahdouz-Mohammadi (roya_kolahdooz@yahoo.com)

Amir-Hassan Zarnani (zarnania@tums.ac.ir)

Version: 1 Date: 15 Sep 2019

Author’s response to reviews:

Dear editor,

Thank you very much for your letter informing us of your decision regarding the manuscript ID: BMWH-D-19-00608 entitled "Evaluation of apoptosis and angiogenesis in ectopic and eutopic stromal cells of patients with endometriosis compared to non-endometriotic controls ". The manuscript has been revised according to your comments. Below are point-by-point responses for your comments. I thank you very much in advance for your effort in this endeavor.

Yours Sincerely,

Amir-Hassan Zarnani, Ph.D, Professor of Immunology

Editor's comments:
1. Can you please amend the figure legends to indicate the precise sample size for each test?
Response: The precise sample size for each test was inserted in figure legends (Page 19) and in sample collection section (page 5).

2. All submissions to BMC Women's Health must have a Declarations section
Response: This section was completed according to BioMed Central's editorial policies (pages 16 and 17)